Author's response to reviews

Title: Community Pharmacists Role in Obesity Treatment in Kuwait: A cross-sectional Study

Authors:

Abdelmoneim Awad (amoneim@hsc.edu.kw)
Mohammad Waheedi (mohdw@hsc.edu.kw)

Version: 4 Date: 27 September 2012

Author's response to reviews:

To: Dr. Jean Adams
BMC-Public Health

27/09/2012
Re: MS: 3912957207368238-Community Pharmacists Role in Obesity Treatment in Kuwait: A cross-sectional Study

Dear Dr. Adams,

We thank you and the reviewer for the comments.

We revised the manuscript in the light of the editor and reviewer’s comments.

Included are:
1. Response to the editor and the reviewer’s comments with indication of changes made in the manuscript.
2. Revised manuscript.

I hope our revision satisfies the reviewers and look forward to hearing from you soon.

Yours sincerely,

Dr. Abdelmoneim Awad
Associate Professor of Clinical Pharmacy
Chairman
Department of Pharmacy Practice
Faculty of Pharmacy
Kuwait University
Editor

1. The authors' qualifications (e.g. PhD / MD) or job titles were removed from the manuscript file. (page 1)
2. The manuscript was revised to conform to the journal style.

Reviewer: Janet Krska

Minor essential revisions

1. Add Kuwaiti to last sentence of abstract background.
Kuwait was added to the last sentence of abstract background (page 2, paragraph 1, line 6).

2. Add indication of meaning of 1 and 5 in abstract method e.g. 1 (low score) to 5 (high score).
Meaning of 1 and 5 was added as being suggested (page 2, paragraph 1, line 8).

3. Conclusion in abstract still does not follow from results presented in abstract and requires to be changed.
Conclusion in abstract was changed to follow from results (page 3, paragraph 2).